

April 12, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 2804-review.doc).

**Title: Response to Abadi and Kusters, *World J Gastroenterol* 9(3): 429-430**

**Author:** Alireza Rafiei, Jeremy J. Gilbreath, and D. Scott Merrell

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 2804

There were no suggested revisions from the reviewer. We have altered the title, running title, and added keywords at the request of the editor. These changes are highlighted in yellow in the submitted "2804-edited" document.

Furthermore, we hereby certify that the language of the manuscript meets or exceeds the "Grade A" requirement as both Dr. Merrell and Dr. Gilbreath are native English speakers with extensive experience publishing in English.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

A. Rafiei, Ph.D.

Molecular and Cell Biology Research Center  
Mazandaran University of Medical Sciences  
Sari 48175-1665, Iran.

Email: rafiei1710@gmail.com